 CHOHAN /26--sept 05

Week 39 starting 26 Sept 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  26 Sept 05 : 9/26/2005 3:16:16 PM  
  
	10
	A1
	MBAL1221
	ICS 1 WP04 HV Test Warm
	296.88

	4
	A2
	MBBR1017
	PT 9 CDAP HV
	1.93

	5
	B1
	SSS642
	PT 2 HV Cold Test
	1.90

	2
	B2
	SSS215
	PT 6.2 Training 2
	3.23 HV

	3
	C1
	MBAL3724
	PT 12 Warm Up
	225.42

	9
	C2
	MBBR3385
	ICS 4 Final connection
	295.8

	1
	D1
	SSS180
	MM 2 LHC Cycle
	1.90

	12
	D2
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	295.89

	8
	E1
	MBAR3394
	Prep 5 Cool Down
	296.25

	6
	E2
	MBBL2203
	Prep 5 Cool Down
	92.05

	11
	F1
	MBAR1252
	Prep 0 Scan Bench
	295.18

	7
	F2
	MBBL2212
	Prep 5 Cool Down
	96.91


  Tempr. from TBSM18 site
   
Coordinator this week: nikolay; MTM support Federico 16-3602

14 complete Magnet Tests cycles 
Only 10 dipoles ( 0 with MM )+ 3 SSS(without MM & SSW ) 

+1 ssss (606 finally after 30 days on bench ) Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· [  B1 WAS EMPTY FOR 5 Days ] ; hope D2 has not the same fate !

Situation From Cryogenics:  J-P Lamboy or ACR Rep.
 Special  Points :
· Richard & Memo HV connector

· Special SSS on D2 : Q9  ( prospects & plans – J Halik ) 
· Power Supply tests for AB-PO (H Thiesen & Giorgio)

	
	AT/MTM


Memo

To:
Operation team
From:
Richard Mompo, Patrick Legrand, Frederic Flamand
Date:
21 Septembre 2005
· New connection on top of CFBs
Frederic Flamand has prepared an installation to ease the connection and disconnection of the cables needed for warm and AC measurements and also to avoid touching the current leads that can be very hot. 
We discovered later on that those cables were causing problems during HV tests at 3100Volts. 

So the electrical insulation of the cables was reinforced, on all 6 clusters, in order to carry out HV tests without having to disconnect those cables anymore! The cables (with brown sleeves cf. picture), between the current leads and the aluminum holds, can remain connected during the whole run of a magnet.
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	Magnets under Test from 18-09-2005 to 24-09-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAL3399 
	[image: image4.png]



	


	14-09-2005 
	30 h.
	23 h.
	19 h.
	26 h.
	11 h.
	19-09-2005 
	109 h.
	Stripping
	
	

	2
	TBB2
	SSS284 
	[image: image6.png]



	


	20-09-2005 
	32 h.
	14 h.
	14 h.
	7 h.
	16 h.
	23-09-2005 
	83 h.
	Stripping
	
	

	3
	TBB2
	SSS343 
	[image: image8.png]



	


	16-09-2005 
	32 h.
	16 h.
	14 h.
	7 h.
	15 h.
	19-09-2005 
	84 h.
	Stripping
	
	

	4
	TBC1
	MBAR2227 
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	19-09-2005 
	19 h.
	26 h.
	18 h.
	12 h.
	12 h.
	23-09-2005 
	87 h.
	Stripping
	
	

	5
	TBC1
	MBBL2214 
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	14-09-2005 
	33 h.
	19 h.
	38 h.
	23 h.
	15 h.
	19-09-2005 
	128 h.
	Stripping
	
	

	6
	TBC2
	MBAR2230 
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	16-09-2005 
	37 h.
	32 h.
	15 h.
	15 h.
	15 h.
	21-09-2005 
	114 h.
	Stripping
	
	

	7
	TBD1
	SSS293 
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	10-09-2005 
	68 h.
	74 h.
	22 h.
	13 h.
	11 h.
	18-09-2005 
	188 h.
	Stripping
	
	

	8
	TBD2
	SSS606 
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	08-09-2005 
	0 h.
	19 h.
	269 h.
	18 h.
	49 h.
	23-09-2005 
	355 h.
	Stand By
	
	

	9
	TBE1
	MBAL1201 
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	17-09-2005 
	28 h.
	31 h.
	13 h.
	13 h.
	7 h.
	20-09-2005 
	92 h.
	Stripping
	
	

	10
	TBE2
	MBBL2201 
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	18-09-2005 
	38 h.
	26 h.
	22 h.
	32 h.
	11 h.
	24-09-2005 
	129 h.
	Stripping
	
	

	11
	TBE2
	MBBL2217 
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	13-09-2005 
	31 h.
	24 h.
	37 h.
	16 h.
	11 h.
	18-09-2005 
	119 h.
	Stripping
	
	

	12
	TBF1
	MBAL1250 
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	17-09-2005 
	32 h.
	22 h.
	14 h.
	18 h.
	19 h.
	22-09-2005 
	105 h.
	Stripping
	
	

	13
	TBF2
	MBAL2134 
	[image: image28.png]



	


	20-09-2005 
	22 h.
	23 h.
	14 h.
	14 h.
	15 h.
	24-09-2005 
	88 h.
	Stripping
	
	

	14
	TBF2
	MBAL3401 
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	16-09-2005 
	28 h.
	24 h.
	17 h.
	14 h.
	10 h.
	20-09-2005 
	93 h.
	Stripping
	
	


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3399
	9636 A.
(D1_U)
	11758 A.
(D1_U)
	12583 A.
(D1_U)
	12845 A.
(D1_L)
	[image: image32.png]



	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS284
	12552 A.
(QE-34)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS343
	12567 A.
(QI-34)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAR2227
	11350 A.
(D1-L)
	12197 A.
(D1_U)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL2214
	10147 A.
(D2_U)
	11996 A.
(D1_L)
	12839 A.
(D2_L)
	12850 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAR2230
	11565.4 A.
(D1_L)
	12150.3 A.
(D1_U)
	12462.2 A.
(D2_U)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS293
	11410 A.
(QI-34)
	11869 A.
(QE-12)
	12278 A.
(QI_12)
	12514.8 A.
(QE_12)
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	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS606
	3968 A.
(MQM-E-2)
	4137 A.
(MQM_E_2)
	4224 A.
(MQM_I_2)
	4248 A.
(MQM-I-2)
	4286 A.
(MQM-I-1)
	4310 A.
(MQM-E-1)
	4324 A.
(MQM-I-1)
	4324.4 A.
(MQM-E-1)
	4351 A.
(MQM- I-1)
	4381 A.
(MQM-E-2)
	4403 A.
(MQM_I_2)
	4400 A.
(MQM-E-1)
	4429.8 A.
(MQM_E_2)
	[image: image39.png]




	9
	MBAL1201
	12643 A.
(D2_L)
	12856 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	10
	MBBL2201
	11623 A.
(D1_U)
	12568 A.
(D2_U)
	12850 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	11
	MBBL2217
	11186 A.
(D1-U)
	12513 A.
(D2-L)
	12857 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	12
	MBAL1250
	12473 A.
(D1_L)
	12630 A.
(D2_U)
	12840 A.
(D2_U)
	[image: image43.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	13
	MBAL2134
	10857 A.
(D2-L)
	12657 A.
(D1_L)
	12848 A.
(D2-U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	14
	MBAL3401
	12290 A.
(D1_L)
	12391 A.
(D2_L)
	12864 A.
(D2_U)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
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Shows Cohabitation D1 – D2 really is a problem !!  SSS on D2 only needed 3 quenches.  D2 is waiting and at cold while D1 is being worked on.
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Thursday_22_M
	TBD1


	Resistance &voltage measurements at warm after cold test: The sextupole voltage taps  have incorrect measurement values.
	MTM
	
	TC informed. NC opened.

On TC's advise the magnet has been signed for stripping as the sextupoles have been trained to the ultimate value of 600A

	Thursday_22_N
	TBC1


	IAP after cold tests on MB2227: signals are not as expected while the same file analysed with hfoff shows correct results
	MTM
	IAP
	Giorgio called. He asked to proceed with the program and will investigate the problem tomorrow.

	Friday_23_N
	TBE


	Cycles on MB1017 aborted due to cryo-loss.
	ACR
	
	

	Saturday_24_M
	TBB2


	Oracle error on TT821 in HWREC when launching IAP at warm. HWREC, files are missing.
	CRI
	
	Giorgio called. The thermometer was changed for this magnet. The coefficients were present in MTF but not loaded by HWREC. He put the HWREC off and created the settings accordingly.

	Saturday_24_M
	TBB2


	Quench heaters did not discharge when power supply were on.
	MTM
	
	Giorgio came. The security matrix was configured for a special SSS. The connector place needed to be changed.

	Saturday_24_M
	TBD1


	Not-correct results SSW measurements on SSS180.
	MTM
	
	Guy called. He came and found that the system is not responding correctly because after switched off once one config script has to be re-launched.

	Saturday_24_M
	TBA1


	Diode test could not be launched. Power supply for LF heater did not charged fully.
	MTM
	
	Giorgio came and solved the problem by changing the capacitor bank.

	Sunday_25_A
	TBD1


	MM1 SSS180. Hardware and configuration fle problem.
	MTM
	
	Giorgio explained the procedure for launching MM1. 

He also came in SM18 because one shaft connector was not correctly plugged. He also uploaded the new coil factor in the DB.

	Sunday_25_N
	TBB2


	On SSS 215 provoked quench 1/2 HF did not show the complete results because of an analysis error:  as the configuration file was not generated correctly.
	MTM
	
	Giorgio advised to proceed further .He completed the config. file Monday morning.

	Monday_26_A
	TBD1


	MM4 (magnetic measurements on corrector) could not be launched. It was not possible to change the power supply from CD to EF
	MTM
	
	The ramp was on 20 A/s in place of 2 A/s


DC cable   and   AC cable 





Permanent connection








